Author's response to reviews

Title: Early childhood caries and its relationship with perinatal, socioeconomic and nutritional risks: a cross-sectional study

Authors:

Valdeci Santos Junior (valdeciodonto@gmail.com)
Rebeca Maria Brasileiro (rebeca-brasileiro@hotmail.com)
Cecilia Maria Couto (cecicouto@hotmail.com)
Aronita Rosenblatt (valdeciodonto@ig.com.br)

Version: 2
Date: 2 January 2014

Author's response to reviews:

The declaration of Interests, Authors' Contributions and Acknowledgements were inserted according to the requests of the editors.

Sincerely,
Dr Valdeci Elias